Author's response to reviews

Title: Valeriana officinalis root extract suppresses physical and psychological stress responses by decreasing the ratio of monoamine neurotransmitters to their metabolites

Authors:

Hyo Young Jung (hyoyoung@snu.ac.kr)
Dae Young Yoo (yo4419@snu.ac.kr)
Woosuk Kim (tank3430@snu.ac.kr)
Sung Min Nam (lovingvet@snu.ac.kr)
Jong Whi Kim (kimjongwhi@snu.ac.kr)
Jung Hoon Choi (jhchoi@kangwon.ac.kr)
Youn-Gil Kwak (dkgline@naturalfnp.com)
Yeo Sung Yoon (ysyoon@snu.ac.kr)
In Koo Hwang (vetmed2@snu.ac.kr)

Version: 2
Date: 14 April 2014

Author's response to reviews:

MS number: 7905659551266067
MS title: Valeriana officinalis root extract suppresses physical and psychological stress responses by decreasing the ratio of monoamine neurotransmitters to their metabolites

Torkel Falkenberg
Editor of BMC Complementary and Alternative Medicine

Apr 14, 2014

Dear Dr. Torkel Falkenberg

Thank you for editorial comments. I inserted the approval number of animal ethics.

Very sincerely yours,

In Koo Hwang, DVM, PhD
Department of Anatomy and Cell Biology
College of Veterinary Medicine, Seoul National University